



**Baishideng
Publishing
Group**

7041 Koll Center Parkway, Suite
160, Pleasanton, CA 94566, USA
Telephone: +1-925-399-1568
E-mail: office@baishideng.com
<https://www.wjgnet.com>

Answering Reviewers

Reviewer 1:

I would like to congratulate the authors for conducting this excellent study. Following reviewing this manuscript, it was found to be excellent. However, one point was found to be more explored in methodology section in both abstract and body of the manuscript in relation to study design and sampling techniques. The remaining parts look good.

Thank you very much for your comments on our manuscript. We have made changes to the methodology section in both abstract and body of the manuscript and marked the modified content in yellow.